



Article The Impact of Corporate Governance on Sustainability Disclosures: A Comparison from the Perspective of Financial and Non-Financial Firms

Asuman Erben Yavuz ¹, Bade Ekim Kocaman ², Mesut Doğan ³, *, Adalet Hazar ², Şenol Babuşcu ², and Raikhan Sutbayeva ⁴

- Vocational School of Social Sciences, Baskent University, 06490 Ankara, Türkiye; aeyavuz@baskent.edu.tr
 International Einance and Banking Baskent University 06490 Ankara, Türkiye;
- ² International Finance and Banking, Baskent University, 06490 Ankara, Türkiye; badeekim@baskent.edu.tr (B.E.K.); ahazar@baskent.edu.tr (A.H.); babuscu@baskent.edu.tr (Ş.B.)
- Banking and Insurance, Bilecik Seyh Edebali University, 11230 Bilecik, Türkiye
- ⁴ Department Management, Yessenov University, 130000 Aktau, Kazakhstan; raikhan.sutbayeva@yu.edu.kz
- * Correspondence: mesutdogan07@gmail.com

Abstract: This study explores the impact of corporate governance on firms' environmental, social, and governance (ESG) performance, with a focus on board characteristics and ownership structures. Using a panel dataset of 6 financial and 16 non-financial firms listed on the Borsa Istanbul (BIST) from 2013 to 2021, the study investigates how ownership (blockholder, foreign, or institutional) and board composition (size, gender diversity, and foreign directors) influence ESG disclosures. The analysis distinguishes between financial and non-financial firms, revealing that corporate governance mechanisms affect ESG performance differently across sectors. Foreign ownership and the presence of foreign and female board members are positively associated with higher ESG disclosures, while ownership concentration is negatively correlated with ESG performance. These findings suggest caution when comparing firms across sectors based solely on ESG disclosures, as governance factors influence outcomes differently in financial and non-financial contexts. This study provides a detailed analysis of effective corporate governance mechanisms in Türkiye, emphasizing the crucial roles of ownership structure and board composition in enhancing ESG transparency. The results offer valuable insights for regulators and investors, contributing to a nuanced understanding of how governance structures shape ESG performance in both financial and non-financial firms in Türkiye.

Keywords: ESG disclosure; corporate governance; boards of directors; ownership structure

1. Introduction

Environmental issues, such as global warming, climate change, and pollution, have wide-ranging effects, disrupting not only ecological balance but also economic and social systems. In response, ESG has emerged as a framework to assess a firm's commitment to environmental responsibility, social equity, and transparent governance. The term "ESG" was first introduced in the 2004 report, *Who Cares Wins—Connecting Financial Markets to a Changing World*, published by the United Nations Finance Initiative [1,2]. This landmark report provided guidelines for investors on incorporating environmental, social, and governance factors into financial assessments to promote long-term financial success. It also marked the beginning of ESG's widespread adoption and clarified the distinctions between sustainability and corporate social responsibility (CSR), ensuring these concepts were not misunderstood or conflated.

CSR has been a key topic of discussion in both the business world and academia since the 1950s. It refers to a firm's voluntary commitment to addressing its responsibilities toward society and the environment [3]. In recent years, sustainability has become increasingly important to businesses, policymakers, and researchers. The United Nations'



Citation: Erben Yavuz, A.; Kocaman, B.E.; Doğan, M.; Hazar, A.; Babuşcu, Ş.; Sutbayeva, R. The Impact of Corporate Governance on Sustainability Disclosures: A Comparison from the Perspective of Financial and Non-Financial Firms. *Sustainability* **2024**, *16*, 8400. https:// doi.org/10.3390/su16198400

Academic Editors: Antonio Prencipe and Danilo Boffa

Received: 14 August 2024 Revised: 18 September 2024 Accepted: 18 September 2024 Published: 27 September 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). Brundtland Report (1987) defines sustainable development as "development that meets the needs of the present without compromising the ability of future generations to meet their own needs" (World Commission on Environment and Development-WCED, 1987) [4]. By the early 2000s, CSR had evolved into the broader framework of ESG, emphasizing that businesses must integrate these factors into their operations to ensure long-term sustainability [5]. CSR is often seen as essential for the economic development of firms [6].

ESG represents a broader and more comprehensive framework than traditional CSR. A firm's ESG score provides a quantitative measure of its CSR performance [7]. ESG enables firms to systematically integrate sustainability and social responsibility into their value-creation processes [8]. As a result, CSR and ESG are often used interchangeably in the literature, reflecting their shared emphasis on balancing corporate responsibility with long-term business success.

ESG factors are increasingly recognized as key drivers of business sustainability and value creation [9]. These factors include a set of policies and practices that help firms evaluate and improve their environmental, social, and governance impacts. By adopting ESG strategies, businesses can mitigate risks, enhance financial performance, and generate value for all stakeholders [10]. Since the launch of the United Nations' Sustainable Development Goals in 2015, awareness of social and environmental issues has expanded, making ESG factors even more important for aligning with sustainable development objectives [11,12]. As ESG gains prominence, firms are encouraged to manage their corporate governance in accordance with social and environmental standards to ensure long-term sustainable growth [13].

Research on the impact of corporate governance on sustainability disclosures offers varied perspectives. Some studies suggest that CSR enhances firm performance [14–18] while others argue that CSR may impede performance [19,20]. Hsiao (2024) [21] notes that shareholders are often concerned about the benefits of social responsibility initiatives, as these efforts can reduce cash flow and potentially lead to financial challenges, such as credit crises, ultimately causing a decline in firm value. In many countries, even the tax advantages associated with CSR fail to significantly enhance profitability or improve overall firm performance.

Building on our exploration of the relationship between specific corporate governance attributes and ESG disclosures, the primary aim of this research is to address the following question: does corporate governance influence ESG disclosures differently in financial versus non-financial firms? By answering this question, our study broadens the conventional understanding of corporate governance and provides initial insight into the potential connection between board diversity, ownership structures, and ESG disclosures. The analysis examines 6 financial and 16 non-financial firms listed on the Borsa Istanbul (BIST) from 2013 to 2021.

What sets this study apart is its comparison of financial and non-financial firms. This distinction is central to the study's structure and is based on the unique characteristics of the two types of firms. First, financial firms, such as banks, act as intermediaries in the economy by channeling funds from savers to borrowers, playing a key role in creating liquidity and credit [22]. In contrast, non-financial firms operate primarily in the real sector, producing goods and services without directly engaging in financial intermediation. The Diamond–Dybvig Model (1983) [23] explains the role of banks as liquidity providers and how this function shapes the relationship between depositors and banks.

Second, financial firms, particularly banks, specialize in money and liquidity management. By accepting deposits and issuing loans, they provide a vital source of liquidity for the economy [24]. Since money, the primary asset of banks, is highly sensitive to economic fluctuations, effective liquidity management is essential. The Keynesian Theory of Liquidity Preference [25] underscores the role of liquidity in the economic system and the factors influencing money demand, highlighting the importance of banks in maintaining economic stability. Finally, financial firms, particularly banks, operate under stringent regulatory frameworks, including capital-adequacy ratios and liquidity requirements, to ensure the stability of the financial system. In contrast, non-financial firms are typically subject to more flexible regulations [26].

This study contributes to the literature in three key areas. First, the existing research on the impact of corporate governance practices on sustainability disclosures presents inconsistent findings. This study addresses these discrepancies by offering a more nuanced analysis of the differences between financial and non-financial firms in Turkey. Second, financial firms face stricter regulatory frameworks compared to other sectors. By examining how these regulations affect corporate governance and ESG disclosures, this study aims to uncover the underlying reasons behind the variations between financial and non-financial firms. The findings will provide valuable insights for regulators and investors, offering strategic recommendations and guidance for policymakers. Lastly, as an emerging market, Türkiye offers a unique context for studying corporate governance and sustainability practices. This study is the first to explore the impact of corporate governance mechanisms on ESG disclosures in Türkiye, contributing significantly to the growing body of ESG literature.

The remainder of the paper is organized as follows. Section 2 provides the theoretical background, and Section 3 presents the literature review and research hypotheses. Section 4 outlines the measurement of the dependent and independent variables, along with the methodology employed. Section 5 presents and discusses the empirical results. Finally, Section 6 offers the conclusions.

2. Theoretical Background

There is no single theory that forms the theoretical foundation of corporate governance; rather, a combination of multiple theories supports discussions and research in this field. These theories collectively contribute to the ethical and effective management of firms. In this section, we discuss six of the most widely accepted corporate governance theories.

Agency Theory, one of the most fundamental theories of corporate governance, was established by Berle and Means (1932) [27] and later expanded by Alchian and Demsetz (1972) [28], Jensen and Meckling (1976) [29], and Fama and Jensen (1983) [30]. Agency Theory addresses the principal–agent relationship within firms, where a board of directors and senior managers act as agents, managing the firm on behalf of the shareholders. The theory focuses on the separation of ownership and management, analyzing the conflicts that arise between the interests of managers and shareholders. It also explores mechanisms to prevent managers from exploiting the firm for personal gain. Agency Theory is the most frequently cited theory in the corporate governance literature and serves as the foundation for many other theories in the field.

Stakeholder Theory, proposed by Freeman (1984) [31], argues that a firm should consider the welfare of all stakeholders, not just shareholders. These stakeholders include employees, customers, suppliers, society, and the environment. The theory suggests that a firm's long-term success and sustainability can only be achieved by balancing and protecting the interests of all stakeholders. From a corporate governance perspective, managers can structure business operations to account for the well-being of all parties involved with the firm.

Legitimacy Theory proposes that firms are viewed as legitimate when they operate in line with the norms and values accepted by the stakeholders and society [32]. A firm's legitimacy influences its reputation and trust among stakeholders and society, which in turn, affects its long-term success. The concept of a social contract between the firm and society holds the firm accountable, as society grants businesses the authority to use resources and employ people [33].

In contrast to Agency Theory, Stewardship Theory views managers as stewards who act in the best interests of shareholders [34]. Managerial and board-member independence is a key aspect of this theory. Independent directors help safeguard shareholders' interests

and enhance the impartiality of management, fostering trust and alignment between managers and shareholders.

The Resource Dependence Theory, introduced by Aldrich and Pfeffer (1976) [35] and further developed by Salancik and Pfeffer (1978) [36], emphasizes the board's relationship with external factors and its role in managing the complexities of the external environment. The board plays a crucial decision-making role in the organization's interactions with external stakeholders [37]. Unlike Agency Theory, which advocates for a predominance of external board members to "ensure control over management", Resource Dependence Theory highlights the "ability of external members to provide the resources needed by the firm". According to this theory, increasing the number of outside directors enables the firm to access more resources, thereby positively impacting firm performance.

Signaling Theory focuses on reducing information asymmetry in corporate governance and minimizing conflicts of interest between managers and [38]. The theory aims to ensure that stakeholders receive accurate and reliable information about the true condition of the firm. When stakeholders lack sufficient information, managers may exploit this asymmetry, potentially leading to moral hazard. Providing clear and credible signals to stakeholders can mitigate the issue of information asymmetry [39].

3. Literature

Studies examining corporate governance in the context of ownership structure and board characteristics in relation to ESG scores focus on how these practices impact ESG performance. These studies consistently emphasize the need for firms to reassess their governance structures to improve ESG outcomes.

The literature linking corporate governance to sustainability performance is primarily based on two dominant theories, namely Agency Theory and Stakeholder Theory. Under Agency Theory, ESG investment is often seen as a proxy for additional costs. In contrast, Stakeholder Theory emphasizes that firms have broader objectives beyond maximizing shareholder value, including fulfilling environmental and social responsibilities that are critical for long-term survival in the market. In the following sections, we present our hypotheses on various board and ownership characteristics.

3.1. Board Size

While there is no consensus on the impact of board size on ESG performance, studies reporting a positive relationship between the number of board members and ESG scores are more prevalent in the literature [40–63]

From a legitimacy perspective, larger boards tend to be more diverse in terms of expertise, experience, and stakeholder representation, which can enhance a company's reputation and image. The inclusion of a wider range of stakeholders on larger boards may also lead to greater involvement in ESG practices [64]. Treepongkaruna et al. (2024) [65] conducted the first study to examine how board governance affects ESG controversy scores, finding that larger boards have a more positive influence on sustainability. This positive relationship is attributed to the presence of experienced and knowledgeable directors on larger boards, who are better equipped to manage environmental issues and provide valuable guidance to the rest of the board.

On the other hand, several studies support Jensen's (1993) [66] claim that smaller boards may be more effective than larger ones due to fewer coordination and communication challenges [12,67–70]. These studies argue that larger boards may struggle with decision-making and controlling management discretion. Research that found a negative relationship between board size and ESG performance suggests that, as the board size increases, ESG factors may be overlooked in decision-making and policy implementation due to communication and coordination difficulties among members.

In addition to the linear effects, some studies suggest a non-linear relationship between board size and ESG performance [71]. This research found that board size exhibits a nonlinear relationship with ESG performance, particularly in non-financial firms, where larger boards tend to demonstrate a stronger commitment to ESG practices.

Although the literature presents mixed findings on the relationship between board size and ESG disclosures, a positive relationship is more frequently emphasized. In light of Türkiye's growing social and environmental initiatives, and in line with the majority of the prior studies, the following hypothesis is proposed.

Hypothesis 1. Board size has a positive effect on ESG disclosures in both financial and non-financial firms.

3.2. Ratio of Female Members in the Board of Directors

Resource Dependence Theory suggests that a gender-diverse board brings varied resources, perspectives, and expertise to address environmental [72]. The presence of more female board members can enhance a firm's sensitivity to social and environmental issues, improve decision-making quality, and positively influence ESG disclosures. As a result, the literature overwhelmingly supports the view that female directors positively impact ESG scores [12,49,50,52,58,61,63,69,70,73–79].

Yadav and Prashar (2023) [80] found that having a small percentage of women directors has minimal impact on ESG performance. However, when there are at least three women on the board, the relationship with ESG performance becomes more favorable. Pinheiro et al. (2023) [59] analyzed the effect of gender diversity on the ESG performances of Latin American companies using both symmetrical (panel data regression) and asymmetrical (fuzzy set qualitative comparative analysis) methods, finding that more gender-diverse boards contribute positively to ESG performance. Similarly, Wasiuzzaman and Subramaniam (2023) [81] found that female directors generally enhance the quality of ESG disclosures and their individual components (excluding governance). However, their analysis of firms in both developed and developing countries revealed that these positive effects are significant only in firms in developed countries.

However, some studies have found a negative relationship between the proportion of female board members and ESG scores [54,82–86]. A key reason cited for this negative relationship is the low number of female members on boards. These studies emphasize that a single female director often cannot exert significant influence and that a minimum of three female board members is necessary for meaningful impact. For female directors to positively affect ESG scores, a critical mass is needed, as a larger group of women on the board can provide new perspectives and contribute positively. When only one woman is present, she may be seen as a "token" and may not effectively offer different viewpoints on board decisions. Additionally, some studies argue that female directors may lack the necessary education or experience to contribute to ESG-related decisions. In sum, these findings suggest that merely having female board members does not automatically lead to diverse perspectives or improved ESG performance.

Deschênes et al. (2015) [87] found that female board members generally had a positive impact on ESG performance, though the relationship was negative when focusing on the environmental dimension. Sepulveda-Nuñez et al. (2024) [71] is the only study to identify a nonlinear relationship between gender diversity and ESG performance.

Although the literature presents mixed findings regarding female board representation and ESG disclosures, most studies suggest a positive relationship between a higher number of female board members and improved ESG scores. Given Türkiye's growing social and environmental initiatives, and in line with the majority of prior research, we propose the following hypothesis.

Hypothesis 2. *The number of female board members positively impacts ESG disclosures in both financial and non-financial firms.*

3.3. Ratio of Foreigners in the Board of Directors

The internationalization of board members is considered beneficial, as it brings diversified expertise, skills, expanded social networks, access to international financing, and opportunities for going public [88]. The inclusion of foreign nationals on a firm's board signals a commitment to enhancing transparency and corporate governance [89].

According to Resource Dependence Theory, foreign board members provide valuable contributions, including diverse viewpoints, cultural perspectives, language skills, and varied life experiences, which can enhance the firm's decision-making process [90]. These perspectives can lead to more well-rounded decisions, ultimately improving the firm's ESG performance.

Several studies have shown that an increase in foreign board members positively impacts ESG disclosures. Lau (2014) [46] found that foreign board members, through their experiences of living, studying, and working in developed economies, help local firms establish higher ethical standards and norms, leading to more transparent disclosures. Similarly, Garanina and Aray (2021) [57] argued that foreign board members bring valuable knowledge and experience, encouraging firms to invest more in ESG practices. The positive relationship is largely attributed to the presence of qualified members with global industry experience and the diverse expertise that foreign board members offer—expertise that local board members may lack. This suggests that foreign board members leverage their international experience to enhance ESG performance.

Toumi et al. (2022) [91] distinguish between two categories of directors—French and their European counterparts—to analyze how they may react differently to ESG factors. The study finds that regional diversity, particularly Anglo-American directors on a board, is positively and significantly correlated with ESG and environmental disclosures. However, this same diversity is negatively correlated with social and governance disclosures.

Some studies have found a negative relationship between the proportion of foreign board members and ESG scores [70,92–94] while others, such as Barako and Brown (2008) [95], reported no significant relationship.

Despite these mixed findings, foreign board members are generally expected to bring positive value to the firm by leveraging their expertise to promote greater involvement in ESG activities. The literature predominantly suggests the potential for a positive relationship between the proportion of foreign board members and ESG scores. Therefore, considering Türkiye's social and environmental initiatives, and in line with most previous studies, the following hypothesis is proposed.

Hypothesis 3. *The proportion of foreign board members has a positive effect on ESG disclosures in both financial and non-financial firms.*

3.4. Blockholder Ownership

Ownership structure refers to the distribution of a firm's ownership among its shareholders, encompassing factors such as the concentration of shares held by individuals or groups of shareholders and whether a single major shareholder controls more than half of the voting capital [96]. Blockholder ownership, where large shareholders hold significant rights and control over the firm, is a critical variable in corporate governance and plays a key role in influencing ESG decisions.

Early studies, examining the relationship between ownership structure and ESG disclosure, indirectly support the hypothesis that ESG levels increase as the firm's ownership becomes more dispersed [97,98]. Keim (1978) [97,98] argue that, as ownership concentration decreases, investors interested in corporate social activities exert more pressure for greater disclosure of social responsibility initiatives. More recent studies, such as those by Brammer and Millington (2005) [99], Li and Zhang (2010) [100], López-Iturriaga and López-de-Foronda (2011) [101], Dam and Scholtens (2013) [102], Rees and Rodionova (2013) [103], and Ellli (2023) [12], found that ownership concentration had a negative effect on ESG scores. These studies suggest that, as controlling shareholders gain more power,

7 of 21

they dominate decision-making, leading to conflicts of interest with other shareholders and reducing the firm's commitment to social responsibility. Additionally, controlling shareholders may view social responsibility projects as non-value-creating, further limiting ESG efforts.

Agency Theory [29] suggests that concentrated ownership structures can reduce conflicts of interest between managers and shareholders. Supporting this theory, studies by Eng and Mak (2003) [104], Harjoto and Jo (2011) [105], Ho and Taylor (2013) [106], and Crisóstomo and Freire (2015) [96] found a positive relationship between block ownership or ownership concentration and ESG scores. These studies argue that large shareholders, who dominate the ownership structure, are motivated by reputational concerns and view ESG policies as a means to enhance the firm's reputation. A concentrated ownership structure may also facilitate better alignment of interests and greater support for ESG initiatives. Additionally, Chung et al. (2019) [107], in their analysis of the Korean market, found that blockholder ownership from previous periods had a notably positive influence on a firm's current corporate governance index. They explain that blockholders are incentivized to maintain and improve the firm's value, as they cannot easily sell their large holdings without negatively impacting their investments.

Given the mixed findings in the literature on the relationship between ownership concentration and ESG disclosures, drawing a definitive conclusion remains challenging. However, considering Türkiye's social and environmental initiatives, the following hypothesis is proposed.

Hypothesis 4. Concentration in the ownership structure has a positive effect on ESG disclosures *in both financial and non-financial firms.*

3.5. Foreign Ownership

The ownership structure is a key determinant of ESG performance [57]. Foreign owners, in particular, tend to be more focused on the quality of corporate governance practices [108] and often push firms to pay higher dividends [109] as they are more likely to align their interests with those of shareholders [110].

According to Resource Dependence Theory, foreign partners with diverse experiences and cultural backgrounds can significantly influence managerial selection and the quality of information provided by the firm [111,112]. Masud et al. (2018) [113] noted that foreign partners play a key role in monitoring and communicating with local and international environmental organizations, acting as a diversified and effective stakeholder group. Delgado-Márquez et al. (2015) [114] further argued that foreign shareholders' concerns about environmental issues often drive local corporate management to comply with environmental regulations and report more sustainable practices to mitigate political risks.

According to Legitimacy Theory, a firm operates within a social contract where societal expectations and corporate interests are intertwined [115]. For a firm's activities to be sustainable, they must be perceived as legitimate by independent stakeholders [116]. Firms that address environmental and social issues often disclose their strategies to gain acceptance and legitimacy from these stakeholder groups [117].

Within the framework of Stakeholder Theory, firms disclose ESG information to align their activities with stakeholders' expectations [95]. By doing so, they strengthen relationships with stakeholders and demonstrate their commitment to socially responsible practices [118].

Several studies suggest that foreign ownership enhances sustainable governance mechanisms and positively influences ESG performance [119,120] with foreign owners acting as motivators for promoting sustainability [113]. Rustam et al. (2019) [51] found that foreign ownership positively impacts ESG performance in Bangladesh, as did Khan et al. (2013) [111] and Zhu et al. (2024) [70] in China and Mishra (2014) [121] in Australia. Fuadah et al. (2022) [122] examined the influence of various ownership structures—foreign, public, state, and family—on ESG disclosure, showing that foreign and public ownership

have a positive and significant impact on ESG disclosure, while state and family ownership do not. Similarly, Ahmed et al. (2024) [123] found that, in non-financial Chinese companies, government and foreign investors significantly improve ESG performance, whereas family investors have a negative effect.

On the other hand, some studies suggest a negative relationship between foreign ownership and ESG performance [124]. For example, Garanina and Aray (2021) [57] found that foreign ownership negatively impacts ESG in Russia, citing limited incentives for foreign ownership and the predominance of non-Western investors, where ESG practices are less developed. Al-Gamrh et al. (2020) [125] observed a similar negative effect in the UAE, and Gulzar et al. (2019) [124] reported the same for China. Haladu and Beri (2016) [126] suggest that the complex relationship between ESG and ownership mechanisms contributes to this negative effect. Additionally, challenges in the foreign-investment process may further complicate this relationship [127,128].

Given the mixed findings in the literature on the relationship between foreign ownership and ESG disclosures, drawing a clear conclusion is difficult. However, considering Türkiye's social and environmental initiatives, the following hypothesis is proposed.

Hypothesis 5. Foreign ownership has a positive effect on ESG disclosures in both financial and non-financial firms.

3.6. Institutional Ownership

The term "institutional ownership" refers to a group of investors who hold significant voting power within a firm. The presence of institutional ownership notably reduces information asymmetry among investors and addresses agency problems [12]. According to the Agency Theory, institutional investors typically demand greater transparency, which is often linked to stronger ESG practices.

In addition to financial performance, institutional shareholders are also concerned with the firm's strategies, activities, and relationships with other [129–133]. As a result, firms with institutional investors are more likely to show greater sensitivity to environmental and social activities, driven by the influence of their corporate partners.

Graves and Waddock (1994) [134], Mahoney and Robert (2007) [133], and Ahmed et al. (2014) [135] found a positive relationship between institutional ownership and ESG performance. Chen et al. (2020) [136] observed that an exogenous increase in institutional holdings enhances the CSR performance of portfolio firms, while Kordsachia et al. (2022) [137] suggested that sustainable institutional ownership is positively linked to a firm's environmental performance. Potharla et al. (2023) [138] also found a significantly positive relationship between persistent institutional ownership and CSR performance in India.

Conversely, some studies report no significant relationship between institutional ownership and ESG. For example, Fauzi et al. (2007) [139] found that institutional investors in Indonesia do not consider ESG in their investment decisions, and Coffey and Fryxell (1991) [140] also found no significant relationship. In contrast, Yadav (2020) [141] identified a negative relationship between ESG performance and institutional ownership in 61 Indian firms from 2013 to 2018. Similarly, Acar et al. (2021) [142], in a study of 27,847 firms across 72 countries or economic regions between 2002 and 2017, found that increased institutional ownership had a negative impact on environmental disclosures.

Given the mixed findings in the literature regarding the relationship between institutional ownership and ESG disclosures, it is difficult to draw a definitive conclusion. However, considering Türkiye's social and environmental initiatives, the following hypothesis is proposed.

Hypothesis 6. *Institutional ownership has a positive impact on ESG disclosures in both financial and non-financial firms.*

The common finding across the studies discussed in this section is that board characteristics and ownership structures significantly influence firms' ESG performance. Recognizing the importance of these factors underscores their key role in improving ESG performance and achieving sustainability goals.

The previous literature has explored the effects of ownership structure and board characteristics on ESG performance, often examining these factors separately. Board characteristics, including size, independence, diversity, and frequency of meetings, have been shown to significantly influence firms' ESG disclosure practices. Similarly, various forms of ownership, such as family, state, foreign, institutional, and blockholder, also play a crucial role in shaping ESG performance. Research suggests that the impact of ownership structure on ESG outcomes is influenced by a range of factors, including market structure, shareholder interests, and local context. The relationship between board and ownership structures and ESG performance is, thus, both complex and multidimensional.

Our study sample comprises both financial and non-financial firms, and we first employed the PCSE regression separately to examine the impact of board and ownership characteristics. We then tested the effects of these variables, including their lagged values, on ESG indicators for financial firms, non-financial firms, and the entire sample. By incorporating a matched sample of financial and non-financial firms, we are able to generalize our findings across both sectors in Türkiye. To the best of our knowledge, this is the first study to empirically investigate the relationship between diverse board and ownership structures and ESG performance in Türkiye, with a detailed distinction between financial and non-financial firms.

4. Data and Methods

The purpose of this study is to examine the impact of corporate governance on firms' sustainability disclosures. Corporate governance is measured by variables related to the board of directors (BoD) and ownership structure, while ESG disclosures are used as indicators of sustainability performance. The study's empirical models are based on data from 6 financial and 16 non-financial firms listed on the Borsa Istanbul (BIST) for the period 2013–2021, with the firms selected based on the availability of ESG scores. ESG data were sourced from the "Refinitiv" website, and additional data were obtained from the firms' annual reports and respective websites. The study focuses on corporate governance variables to assess their influence on sustainability disclosures. The variables used are outlined in Table 1, following the framework of [11].

Variables	Abbr.	Measurement								
Dependent Variable										
ESG Score	ESG	This score measures the extent of ESG information disclosed by firms (f), ranging from 0 to 100. A score of 0 indicates no ESG disclosure, while a score of 100 signifies complete ESG disclosure.								
	Inde	ependent Variables								
Institutional Investors	INST	The ratio of shares held by institutional investors								
Foreign Investors	FOREIGN	The ratio of shares held by foreign investors								
Blockholder	BLOCK	The ratio of shares held by the largest block								
Ownership	DLUCK	shareholders								
Board Size	BSIZE	Number of board members								
Foreign Directors	BFOR	Ratio of foreign directors in BoD								
Female Directors	BFEM	Ratio of female directors in BoD								

Table 1. ESG, corporate governance, and control variables.

Variables	Abbr.	Measurement							
	Control Variables								
Firm Size	SIZE	Logarithm of Total Assets							
Leverage Ratio	LEV	Total Debt/Total Assets							
Return on Assets	ROA	Net Profits/Total Assets							

The following model was developed using the dependent and independent variables from the study, based on Ellili (2023) [12]:

$$ESG_{i,t} = \beta_0 + \beta_1 INST_{i,t} + \beta_2 FOREIGN_{i,t} + \beta_3 BLOCK + \beta_4 BSIZE + \beta_5 BFOR + \beta_6 BFEM + \beta_7 SIZE + \beta_8 LEV + \beta_9 ROA + \varepsilon_{i,t}$$
(1)

The study utilizes the panel-corrected standard errors (PCSE) method, which effectively addresses common challenges in panel data analysis, such as heteroskedasticity and autocorrelation. PCSE accounts for correlations across entities within a panel, enhancing the accuracy of standard error estimates. It corrects for heteroskedasticity, where error variances differ across entities, and adjusts for autocorrelation, ensuring reliable estimates over time. These features make PCSE particularly valuable for obtaining robust inferences in empirical models with panel data. Its applicability to large and complex panel data sets, combined with its flexibility across various data structures, makes it a preferred method for rigorous panel data analysis.

5. Findings

5.1. Descriptive Statistics and Correlation Analysis

Table 2 presents the descriptive statistics for the dependent and independent variables used in the study.

Variables —	Mean				Std. Dev.			Min			Max		
vallables	All	Financial	Other										
ESG	59.38	60.78	58.86	18.69	14.03	20.18	15.36	24.57	15.36	94.18	91.28	94.18	
INST	56.96	58.08	56.54	18.73	15.32	19.89	0.00	37.08	0.00	85.00	83.86	85.00	
FOREIGN	11.05	13.01	10.32	17.27	18.98	16.59	0.00	0.00	0.00	55.00	49.85	55.00	
BLOCK	43.33	45.00	42.71	12.17	8.43	13.28	14.07	25.01	14.07	75.30	75.30	67.54	
BSIZE	10.18	10.65	10.01	2.78	1.89	3.03	5.00	7.00	5.00	18.00	17.00	18.00	
BFOR	16.26	14.16	17.05	19.02	21.18	18.16	0.00	0.00	0.00	54.55	50.00	54.55	
BFEM	11.25	11.95	10.98	11.59	6.96	12.91	0.00	0.00	0.00	45.45	27.27	45.45	
SIZE	7.79	8.54	7.51	0.62	0.23	0.48	6.43	8.15	6.43	9.05	9.05	9.01	
LEV	0.68	0.90	0.59	0.21	0.02	0.18	0.20	0.86	0.20	0.95	0.95	0.88	
ROA	0.04	0.01	0.06	0.04	0.01	0.04	-0.07	0.00	-0.07	0.15	0.02	0.15	

Table 2. Descriptive statistics for financial and non-financial firms.

The analysis shows that the average ESG ratios, as well as the ratios of institutional, foreign, and concentrated investors, and the mean number of board members, female directors, and foreign managers, are comparable between financial and non-financial firms. However, financial firms exhibit larger asset sizes than non-financial firms when reviewing the control variables. Significant differences are also noted in the leverage and return on assets ratios, reflecting the distinct financial structures of these two types of firms. Table 3 highlights the relationship between corporate governance variables and ESG scores for both financial and non-financial firms.

The results show a positive relationship between ESG and board size, the ratio of female board members, and the ratio of foreign board members. There is also a weak negative relationship between ESG and ownership concentration, along with a positive relationship between firm size, represented by total assets, and ESG. Lastly, the absence of a high correlation among the independent variables enhances the model's reliability.

				5						
	ESG	INST	FOREIGN	BLOCK	BSIZE	BFOR	BFEM	SIZE	LEV	ROA
ESG	1									
INST	0.0237	1								
FOREIGN	0.0172	0.3351	1							
BLOCK	-0.0926	0.594	-0.007	1						
BSIZE	0.3136	0.3634	0.285	0.1147	1					
BFOR	0.214	-0.1381	0.5983	-0.486	0.1839	1				
BFEM	0.3377	-0.2256	-0.0794	-0.1729	0.0991	0.2589	1			
SIZE	0.3366	-0.1109	-0.1283	-0.0012	0.1028	-0.037	0.4263	1		
LEV	0.2066	-0.0567	0.1842	-0.174	0.3028	0.2545	0.3029	0.632	1	
ROA	-0.1312	-0.1231	-0.1166	-0.1139	-0.2354	0.0385	-0.2779	-0.5784	-0.5134	1

	Table 3.	Correl	lation	anal	vsis
--	----------	--------	--------	------	------

5.2. Ownership Structure and ESG Disclosures

In the panel data analysis, the Hausman test is used to determine whether a fixedeffects or random-effects model is more appropriate. The test results indicate that the random-effects model is more efficient for this analysis. In the random-effects model, crosssectional effects are assumed to follow a random distribution. Additionally, the variance inflation factor (VIF) and the Breusch–Pagan tests confirm the absence of multicollinearity and heteroskedasticity, respectively.

Four models were developed to examine the relationship between ownership structure and ESG disclosure scores for both financial and non-financial firms. In these models, the impact of each ownership-structure variable on ESG disclosures is first analyzed individually, followed by an assessment of the combined effect of all ownership-structure variables. The results are presented in Table 4.

Table 4. The relationship between ownership structure and ESG disclosures.

		All	Firms		Financial Firms				Non-Financial Firms			
	1	2	3	4	5	6	7	8	9	10	11	12
D IOT	0.101 ***			0.196 **	0.0330			-0.1127 *	0.223 ***			0.435 ***
INST	(1.51)	-	-	(2.17)	(0.46)	-	-	(1.75)	(2.68)	-	-	(4.05)
FOREIGN		0.097		0.036	. ,	0.1583 ***		0.1744 ***	. ,	0.047		-0.137
FOREIGN	-	(1.29)	-	(0.42)	-	(3.75)	-	(4.49)	-	(0.40)	-	(-0.96)
DI OCIV			-0.132	-0.317 ***			-0.4478 ***	-0.4178 ***			0.054	-0.394 ***
BLOCK	-	-	(0.50)	(-2.79)	-	-	(-3.48)	(-3.46)	-	-	(0.50)	(-3.02)
SIZE	12.42 ***	12.84 ***	11.71 ***	13.95 ***	53.214 ***	51.832 ***	54.345 ***	51.150 ***	18.81 ***	17.08 ***	16.40 ***	17.80 ***
SIZE	(4.25)	(4.13)	(3.94)	(4.61)	(14.18)	(15.54)	(16.68)	(15.40)	(6.26)	(4.45)	(5.12)	(5.18)
1 121/	3.35	-1.93	-1.47	-5.64	-321.64 ***	-264.66 ***	-253.90 ***	-162.55 **	21.92 **	17.31 *	20.25 *	18.10 *
LEV	(0.35)	(-0.20)	(-0.15)	(-0.56)	(-4.07)	(-3.93)	(-4.27)	(-2.56)	(2.31)	(1.68)	(1.80)	(1.66)
DOA	57.54	54.57	36.08	52.11	-5.939	-34.056	22.012	-49.905	53.38	29.68	29.36	38.52
ROA	(1.46)	(1.43)	(0.95)	(1.32)	(-0.02)	(-0.16)	(0.11)	(-0.27)	(1.35)	(0.75)	(0.77)	(0.93)
Wald chi2	40.37	33.67	31.08	43.43	260.30	283.64	292.93	322.06	65.25	41.98	44.96	78.67
Prob > chi2	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000

Note: *, **, and *** indicate significance at 10%, 5%, and 1% levels, respectively.

The analysis reveals that the impact of the ownership structure variables on ESG disclosures differs between financial and non-financial firms. For financial firms, an increase in foreign ownership positively influences ESG scores, while the effect of foreign ownership on non-financial firms and the overall sample is less clear. Additionally, an increase in institutional investor ownership decreases ESG scores for financial firms but increases them for non-financial firms. Across all models and sectors, a higher ownership concentration consistently leads to lower ESG scores. In summary, as firms diversify their ownership structures and move away from being family-owned or dominated by large shareholders, such as the public sector, their ESG performance improves.

The analysis of the control variables shows a positive relationship between firm size and ESG disclosure across all sectors, likely due to the greater willingness of larger firms to invest in social and environmental projects. In addition, larger firms face more pressure to engage in ESG activities. However, profitability ratios do not appear to influence ESG disclosures in any sector. Lastly, the ESG disclosure scores increase as the leverage ratios decrease for financial firms. Specifically, financial firms with stronger equity capital and lower risk tend to be more transparent in demonstrating their compliance with social and environmental initiatives.

5.3. Board Structure and ESG Disclosures

In this section, models are developed using three different samples, namely financial firms, non-financial firms, and all firms combined. The empirical analysis first examines the individual impact of each board-structure variable on ESG disclosures, followed by an assessment of the combined effect of all board-structure variables. Table 5 presents the results, focusing on the influence of board size, female board ratio, and foreign board ratio on firms' sustainability performance. The results show a positive relationship between board size and ESG disclosures for both the entire sample and the non-financial firms.

		All F	irms		Financial Firms				Non-Financial Firms			
	1	2	3	4	5	6	7	8	9	10	11	12
BSIZE	2.406 *** (5.76)	-	-	2.084 *** (5.31)	-0.447 (-1.12)	-	-	-0.985 ** (-2.20)	2.370 *** (5.36)	-	-	2.376 ** (5.31)
BFOR	-	0.196 ** (2.40)	-	0.168 *** (2.43)	-	0.1039 *** (2.73)	-	0.132 *** (3.15)	-	0.196 ** (2.40)	-	0.190 *
BFEM	-	-	0.213 *** (3.75)	0.277 ** (2.54)	-	-	0.166 (1.46)	0.059 (0.52)	-	-	0.115 (0.74)	0.061 (0.42)
SIZE	14.083 *** (5.20)	16.741 *** (5.06)	13.429 *** (4.49)	13.083 *** (4.59)	54.464 *** (13.88)	54.797 *** (14.74)	52.432 *** (12.70)	54.957 *** (14.43)	18.117 *** (6.31)	16.741 *** (5.06)	14.597 *** (3.67)	17.707 * (4.97)
LEV	-8.487 (-1.00)	7.509 (0.51)	-6.90 (-0.72)	-17.04 ** (-1.94)	-243.44 *** (-4.21)	-209.45 *** (-3.82)	-216.00 *** (-3.65)	-199.66 *** (-3.73)	5.709 (0.61)	7.509 (0.51)	16.806 * (1.65)	-6.21
ROA	74.809 ** (2.02)	13.859 (0.38)	29.705 (0.80)	65.344 * (1.86)	-140.159 (-0.71)	-105.132 (-0.58)	-119.47 (-0.60)	-124.596 (-0.73)	58.037 (1.56)	13.859 (0.38)	27.617 (0.76)	-48.23 (1.32)
Wald chi2 Prob> chi2	82.53 0.0000	42.83	42.91 0.0000	94.53 0.0000	223.25	243.56 0.0000	219.65	261.27 0.0000	103.15	42.83	40.66	101.9

Table 5. The relationship between board structure and ESG disclosures.

Note: *, **, and *** indicate significance at 10%, 5%, and 1% levels, respectively.

According to Pfeffer (1973) [143], as the size of the board increases, the organization gains more experience and valuable resources. Dalton, Daily, and Johnson (1999) [144] similarly argued that larger boards provide greater knowledge, experience, and improved advisory capabilities. However, Larmou and Vafeas (2009) [145] noted that it can be more difficult to exert managerial pressure on larger boards. Despite these insights, the impact of board size on ESG disclosure varies for financial firms, where a smaller board size is associated with improved ESG performance.

Another key finding of the study is that foreign board members have a positive impact on ESG performance across all sectors. According to Ellili (2023) [12], foreign board members focus not only on financial performance but also on the interests of stakeholders. These results align with the findings of Al Maeeni et al. (2024) [61] and Ellili (2023) [12]. Additionally, there is a positive relationship between female board members—another indicator of board diversity—and ESG performance for the overall sample. However, the effect of female board members on ESG disclosures is not statistically significant when analyzed separately for financial and non-financial firms. Despite this, the significant positive effect of female board members on ESG performance for the overall sample, along with the positive coefficients for both financial and non-financial firms, remains an important finding.

Regarding the control variables, total asset size has a positive relationship with ESG disclosures across the entire sample. However, the return on assets does not significantly affect ESG performance. Additionally, ESG transparency tends to increase as financial firms exhibit higher leverage.

5.4. Corporate Governance and ESG Disclosures

In this section, the effects of corporate governance variables and their lagged values on ESG disclosures are tested separately for financial and non-financial firms, and the entire sample, with the corresponding models developed. The empirical results on the impact of corporate governance practices on ESG disclosures are presented in Table 6.

	All	Financial	Non-Financial		All	Financial	Non-Financial
INICT	0.130 *	-0.269 ***	0.311 ***		0.143 *	0.113	0.343 ***
INST	(1.66)	(-4.23)	(2.68)	INST $t - 1$	(1.67)	(0.85)	(2.91)
FOREIGN	-0.152	0.335 ***	-0.033	FOREIGN t	0.352 ***	0.579 **	0.512 ***
FUREIGN	(-1.54)	(2.40)	(-0.34)	- 1	(3.22)	(2.09)	(3.78)
BLOCK	-0.120	-0.046	-0.125	BLOCK t – 1	-0.047	0.117	-0.175
BLOCK	(-1.06)	(-0.55)	(-0.90)	BLOCK I = I	(-0.37)	(0.61)	(-1.11)
BSIZE	1.978 ***	-1.494 ***	1.953 ***	BSIZE t – 1	1.457 ***	-1.373 *	1.406 **
DSIZE	(4.41)	(-3.60)	(3.41)	DSIZE I - I	(3.15)	(-1.65)	(2.52)
BFOR	0.237 ***	0.253 **	0.342 **		0.391 ***	0.264 **	0.488 ***
	(2.66)	(2.55)	(3.43)	BFOR $t - 1$	(3.97)	(2.22)	(4.30)
BFEM	0.251 ***	0.065	0.021	BFEM $t - 1$	0.166	-0.002	0.124
DFEIVI	(2.24)	(0.61)	(0.15)	DFEIVI $t = 1$	(1.36)	(-0.001)	(0.77)
CLZE	12.802 ***	50.799 ***	15.210 ***		3.730	33.122 ***	1.998
SIZE	(4.43)	(13.96)	(3.80)	SIZE $t - 1$	(1.24)	(4.49)	(0.47)
	-16.520 *	-57.418	-0.033		-4.857	-581.859	-2.445
LEV	(-1.85)	(-1.03)	(-0.00)	LEV $t - 1$	(-0.52)	***	(-0.21)
						(-5.28)	
ROA	54.73	240.10	30.104	ROA t – 1	-13.751	-351.18	-27.616
non	(1.48)	(-1.59)	(0.76)	NOTIC 1	(-0.36)	(-1.20)	(0.70)
Wald chi2	101.56	358.88	120.75	Wald chi2	65.73	65.73	68.89
Prob > chi2	0.0000	0.0000	0.0000	Prob > chi2	0.0000	0.0000	0.0000

Table 6. The relationship between corporate governance and ESG disclosures.

Note: *, **, and *** indicate significance at 10%, 5%, and 1% levels, respectively.

The results show that institutional ownership decreases ESG performance for financial firms, while the effect of the one-year lagged corporate governance variable on ESG performance is insignificant. However, for non-financial firms and the entire sample, institutional investor ownership contributes to an increase in ESG performance. Additionally, foreign investor ownership positively influences ESG disclosures for financial firms. Similarly, for the entire sample, as well as for both financial and non-financial firms, the one-year lagged value of foreign investor ownership significantly boosts ESG performance. This may be because foreign investors in developing countries often become controlling shareholders in regions with weak corporate governance mechanisms, helping to fill the regulatory gap [146].

There is a negative relationship between the number of board members, including its lagged variable, and the ESG performance of financial firms. This may be because, as the number of board members increases, communication and coordination issues arise, management control weakens, and agency problems emerge due to the separation of management and oversight. Conversely, for the entire sample and the non-financial firms, an increase in board size, along with its lagged value, positively affects ESG performance. This is explained by the RDT, which suggests that a larger board provides diverse expertise and resources to the firm.

Another important finding is that an increase in the ratio of foreign board members, as well as its lagged value, has a significant positive effect on ESG disclosures across all industry groups. This aligns with Agency Theory (AT), which posits that diversity on the board, particularly in terms of ethnicity, enhances the board's independence. In other words, greater international diversity on the board broadens the range of resources available to the firm.

Regarding the control variables, firm size positively influences ESG performance, although the effect diminishes as the lagged value of firm size begins to decrease its impact. Similarly, the effect of the lagged leverage ratio on ESG becomes more uncertain. Profitability, on the other hand, does not significantly impact ESG performance in any scenario.

5.5. Corporate Governance and Individual ESG Disclosures

This section examines the effects of all the corporate governance variables, including ownership structure, board characteristics, and control variables, on individual ESG disclosures. The relationship between corporate governance variables and the individual components of ESG, namely environmental, social, and governance, is presented in Table 7.

Table 7. The relationship between corporate governance and individual ESG disclosures.

		Environmenta	al		Social			Governance	
	All	Financial	Non- Financial	All	Financial	Non- Financial	All	Financial	Non- Financial
INICT	0.102	-0.529 ***	0.523 ***	0.107	0.068	0.111	-0.023	-0.561 ***	0.234 *
INST	(1.00)	(-4.16)	(3.84)	(1.23)	(0.69)	(0.88)	(-0.25)	(-3.82)	(1.89)
FODEICN	0.063	0.363	0.271 *	0.191 *	-0.048	0.270 *	0.229 **	0.220	0.425 ***
FOREIGN	(0.50)	(0.165)	(1.76)	(1.73)	(-0.28)	(1.81)	(2.19)	(0.92)	(3.05)
BLOCK	-0.057	-0.278	-0.270	-0.118	0.112	-0.171	0.046	-0.193	0.012
DLUCK	(-0.41)	(-0.97)	(-1.54)	(-1.03)	(0.81)	(-1.14)	(0.32)	(-0.85)	(0.07)
BSIZE	1.818 ***	-0.704	1.131 *	2.334 ***	-0.969	2.606 ***	-0.088	-3.330 ***	-0.509
	(3.26)	(-3.60)	(1.66)	(4.23)	(-1.50)	(3.91)	(-0.19)	(-3.58)	(-0.98)
BFOR	0.113	-0.006	0.321 **	0.203 **	0.336 **	0.122	0.317 ***	0.346	0.556 ***
DIOK	(1.02)	(-0.03)	(2.47)	(2.08)	(2.01)	(1.08)	(3.15)	(1.49)	(5.53)
BFEM	0.353 ***	0.427 **	0.270 *	0.024	-0.228	-0.135	0.359 ***	0.568 **	0.170
DFENI	(2.86)	(2.00)	(1.68)	(0.21)	(-1.25)	(-0.87)	(2.72)	(2.28)	(1.08)
SIZE	15.824 ***	40.338 ***	12.574 ***	22.165 ***	59.419 ***	23.769 ***	-6.005 ***	34.646 ***	-5.621
SIZE	(4.57)	(5.33)	(2.66)	(6.92)	(10.46)	(5.30)	(-2.11)	(3.92)	(-1.49)
LEV	-3.827	-41.393	2.337	-13.034	-76.759	2.664	-7.682	9.177	0.043
LEV	(-0.36)	(-0.35)	(0.18)	(-1.28)	(-1.04)	(0.21)	(-0.93)	(0.07)	(0.00)
ROA	86.55 **	142.44	54.781	7.49	-338.49	-10.57	81.67 **	-188.41	52.105
NOA	(2.06)	(0.50)	(1.27)	(0.18)	(-1.63)	(-0.25)	(1.96)	(-1.63)	(1.25)
Wald chi2	109.35	181.14	120.75	197.24	181.66	146.57	67.76	89.46	57.25
Prob> chi2	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000

Note: *, **, and *** indicate significance at 10%, 5%, and 1% levels, respectively.

The results indicate a negative relationship between the environmental and governance dimensions of ESG and institutional investor ownership in financial firms, while no significant relationship is observed for the social dimension. In contrast, for non-financial firms, an increase in institutional investor ownership improves performance in the environmental and governance dimensions but has no significant effect on the social dimension.

For the entire sample and non-financial firms, there is a positive relationship between board size and environmental and social disclosures. However, in financial firms, board size has a negative relationship with governance disclosures. Consistent with the total ESG score, the proportion of foreign board members positively influences ESG disclosures. One of the most notable findings is that an increase in the proportion of female board members contributes positively to the environmental dimension of ESG across all sectors. Firm size, a control variable, also shows a positive relationship with ESG disclosures, similar to the total ESG score. Unlike the overall ESG results, an increase in return on assets for the entire sample leads to an improved environmental performance.

6. Conclusions and Recommendations

This study offers several theoretical, practical, social, and environmental implications within the context of the corporate governance and ESG disclosure literature, which are particularly relevant for various stakeholders in Türkiye.

The answers to our research question, which form the core motivation of this study, are significant from multiple perspectives. We find that sector-specific regulations have varying effects on sustainability disclosures through corporate governance practices. Specifically, our empirical results show that the impacts of institutional investor ownership and board size differ between non-financial and financial firms. These findings help fill an important gap in the literature, particularly concerning the differing effects of corporate governance on ESG disclosures across sectors.

The first key finding of our study explains why the impact of institutional investor ownership on ESG disclosures varies across sectors, as observed in the literature. In financial firms, an increase in institutional ownership decreases ESG performance, while in non-financial firms, institutional investor ownership improves ESG performance. This difference arises because financial firms in Türkiye are subject to strict governance regulations and are required to make corporate decisions independently of their ownership structure due to rigorous sector oversight and auditing systems. The negative impact of increased institutional ownership on financial firms may be due to the stability of policies implemented by qualified individual shareholders. Changes in decision-makers, resulting from shifts in the ownership structure, can lead to policy changes that negatively affect ESG performance.

In contrast, non-financial firms are governed by general legislation and auditing rules, where investors' decisions have a more direct influence on the firm's operations. Since individual shareholders are not legally required to meet specific qualifications, institutional shareholders may be better equipped to make informed decisions. This explains why institutional ownership of non-financial firms positively impacts ESG disclosures, while non-institutional investors tend to have a negative effect.

The second key finding highlights the differing relationship between board size and ESG performance in financial and non-financial firms. Similar to ownership structure, sector-specific regulations of financial firms clearly define board-member qualifications and set a minimum number of members based on the corporate structure's needs. In this context, financial firms that already meet the required minimum number of qualified board members do not necessarily benefit from adding more members to improve decision-making. As a result, the relationship between board size and ESG disclosures for financial firms is found to be negative.

In contrast, for non-financial firms, an increase in board size leads to improved ESG performance. This is because the managers of these firms tend to act more independently, making decisions that are less constrained by sector-specific regulations and economic requirements. In such cases, corporate shareholders' decisions may align more closely with market demands, positively influencing ESG performance.

The differences in the impact of ownership structure and board composition on ESG disclosures between financial and non-financial firms can be attributed to the unique characteristics of these industries and their distinct regulatory and operational environments. Financial firms operate under stricter regulatory frameworks, particularly regarding risk management, transparency, and governance. This regulatory pressure often results in more conservative, risk-averse behavior, especially regarding ESG disclosures. For instance, the positive effect of foreign ownership on ESG scores in financial firms may be driven by the need to comply with international standards.

Moreover, the core activities of financial firms are closely tied to public trust and risk management, making them more sensitive to changes in ownership and governance structures that could affect their reputation and risk profile. Financial firms also engage with a broader range of stakeholders, including regulators, investors, and customers, who may demand more comprehensive ESG disclosures. These industry-specific dynamics highlight the importance of considering sector-specific factors when analyzing the impact of ownership and governance structures on ESG disclosures.

The empirical findings indicate that foreign ownership, the presence of foreign board members, and a higher number of female board members significantly enhance ESG disclosures in both financial and non-financial firms. These results align with prior research demonstrating the positive impact of cultural and gender diversity on firm performance. In contrast, ownership concentration negatively affects ESG transparency, highlighting the importance of inclusive decision-making. To improve ESG reporting, firms should prioritize diversifying both ownership structures and board composition. These insights provide a valuable framework for strengthening corporate governance practices and expanding ESG disclosures.

Policymakers, particularly in developing countries with weaker corporate governance mechanisms, may consider introducing incentives to encourage foreign investors to take more active roles. Attracting foreign investors can help improve firms' overall ESG performance and enhance the country's reputation in global markets. Additionally, the positive impact of board diversity on ESG disclosures suggests that policies promoting greater diversity on corporate boards should be developed. This could include mandating minimum gender diversity requirements or encouraging the inclusion of board members with international experience. Furthermore, the findings highlight that different ownership structures influence ESG disclosures differently across sectors. Policymakers should account for these sectoral differences when designing corporate governance regulations and guidelines.

In light of the findings from our study, regulatory authorities and investors should exercise caution when evaluating corporate ownership and board composition across financial and non-financial firms. Specifically, ESG disclosures should not be used as a comparative factor between financial and non-financial sector firms, as the nature of these industries differs significantly. However, ESG disclosures can provide meaningful insights when comparing firms within the financial sector or non-financial firms.

This study, which explores the relationship between corporate governance and ESG performance, has certain limitations. The small sample size, limited to firms with ESG scores listed on the Borsa Istanbul, restricts the generalizability of the findings. Furthermore, while six corporate governance variables are examined, other important factors, such as board independence and executive compensation, remain unaddressed. Future research could bridge these gaps by incorporating a broader range of governance variables and expanding the sample to include firms from both developed and developing countries. Moreover, the use of advanced methodologies, such as dynamic panel models, could offer deeper insights into causal relationships. Cross-country and longitudinal studies would further enhance the understanding of how corporate governance influences ESG disclosures across different regions and sectors, ultimately contributing to a more comprehensive understanding of the drivers behind ESG performance.

Author Contributions: Conceptualization, A.E.Y. and B.E.K.; methodology, M.D.; software, M.D.; validation, A.H., R.S. and Ş.B.; formal analysis, M.D.; investigation, A.E.Y.; resources, B.E.K.; data curation, M.D.; writing—original draft preparation, A.H.; writing—review and editing, Ş.B.; visualization, R.S.; supervision, A.H.; project administration, Ş.B.; funding acquisition, A.H. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: The data sets used for experiments and analysis are public and available.

Conflicts of Interest: The authors declare no conflicts of interest.

References

- United Nations Global Compact. Who Cares Wins—Connecting Financial Markets to a Changing World. 2004. Available online: https://www.unepfi.org/fileadmin/events/2004/stocks/who_cares_wins_global_compact_2004.pdf (accessed on 10 July 2023).
- Saharti, M.; Chaudhry, S.M.; Pekar, V.; Bajoori, E. Environmental, social and governance (ESG) performance of firms in the era of geopolitical conflicts. *J. Environ. Manag.* 2024, 351, 119744. [CrossRef] [PubMed]
- 3. Carroll, A.B. Corporate social responsibility: Evolution of a definitional construct. Bus. Soc. 1999, 38, 268–295. [CrossRef]
- World Commission on Environment and Development (WCED). Energy: The Power to Develop 1987; Volume 17, pp. 1–91. Available online: https://idl-bnc-idrc.dspacedirect.org/bitstream/handle/10625/152/WCED_v17_doc149.pdf (accessed on 2 July 2024).
- Eccles, R.G.; Ioannou, I.; Serafeim, G. The impact of corporate sustainability on organizational processes and performance. *Manag. Sci.* 2014, 60, 2835–2857. [CrossRef]
- 6. Škare, M.; Golja, T. The impact of government CSR supporting policies on economic growth. *J. Policy Model.* **2014**, *36*, 562–577. [CrossRef]

- Yoon, B.; Lee, J.H.; Byun, R. Does ESG performance enhance firm value? Evidence from Korea. Sustainability 2018, 10, 3635. [CrossRef]
- 8. Schaltegger, S.; Wagner, M. Integrative management of sustainability performance, measurement and reporting. *Int. J. Account. Audit. Perform. Eval.* **2006**, *3*, 1–19. [CrossRef]
- 9. Friede, G.; Busch, T.; Bassen, A. ESG and financial performance: Aggregated evidence from more than 2000 empirical studies. *J. Sustain. Financ. Investig.* **2015**, *5*, 210–233. [CrossRef]
- Clark, G.L.; Feiner, A.; Viehs, M. From the Stockholder to the Stakeholder: How Sustainability Can Drive Financial Outperformance. Available online: https://papers.ssrn.com/sol3/papers.cfm?abstract_id=2508281 (accessed on 10 July 2023).
- 11. Khan, M.; Serafeim, G.; Yoon, A. Corporate sustainability: First evidence on materiality. *Account. Rev.* **2016**, *91*, 1697–1724. [CrossRef]
- 12. Ellili, N.O.D. Impact of corporate governance on environmental, social, and governance disclosure: Any difference between financial and non-financial companies? *Corp. Soc. Responsib. Environ. Manag.* **2023**, *30*, 858–873. [CrossRef]
- 13. Kamal, Y. Stakeholders expectations for CSR-related corporate governance disclosure: Evidence from a developing country. *Asian Rev. Account.* **2021**, *29*, 97–127. [CrossRef]
- McWilliams, A.; Siegel, D.S.; Wright, P.M. Corporate social responsibility: Strategic implications. J. Manag. Stud. 2006, 43, 1–18. [CrossRef]
- 15. Flammer, C. Corporate social responsibility and shareholder reaction: The environmental awareness of investors. *Acad. Manag. J.* **2013**, *56*, 758–781. [CrossRef]
- Cheng, B.; Ioannou, I.; Serafeim, G. Corporate social responsibility and access to finance. *Strateg. Manag. J.* 2014, 35, 1–23. [CrossRef]
- 17. Al-Shammari, M.A.; Banerjee, S.N.; Rasheed, A.A. Corporate social responsibility and firm performance: A theory of dual responsibility. *Manag. Decis.* 2022, 60, 1513–1540. [CrossRef]
- 18. Liu, L.; Ren, H.; Chen, T.; Chen, J. Research on Impact of Internet Corporate Social Responsibility on Corporate Financial Performance Based on Linear Regression Model. *J. Financ. Account.* **2023**, *11*, 26. [CrossRef]
- 19. Wagner, M.; Schaltegger, S. The effect of corporate environmental strategy choice and environmental performance on competitiveness and economic performance:: An empirical study of EU manufacturing. *Eur. Manag. J.* 2004, 22, 557–572. [CrossRef]
- 20. Moore, G. Corporate social and financial performance: An investigation in the UK supermarket industry. *J. Bus. Ethics* **2001**, *34*, 299–315. [CrossRef]
- 21. Hsiao, H.F.; Zhong, T.; Wang, J. Does national culture influence corporate social responsibility on firm performance? *Humanit. Soc. Sci. Commun.* **2024**, *11*, 1–9. [CrossRef]
- 22. Allen, F.; Santomero, A.M. The Theory of Financial Intermediation. J. Bank. Financ. 1997, 21, 1461–1485. [CrossRef]
- 23. Diamond, D.W.; Dybvig, P.H. Bank runs, deposit insurance, and liquidity. J. Political Econ. 1983, 91, 401-419. [CrossRef]
- 24. Kashyap, A.K.; Rajan, R.G.; Stein, J.C. Banks as Liquidity Providers: An Explanation for the Coexistence of Lending and Deposit-Taking. *J. Financ.* 2002, *57*, 33–73. [CrossRef]
- 25. Keynes, J.M. The General Theory of Employment, Interest and Money; Macmillan: New York, NY, USA, 1936.
- 26. Barth, J.R.; Caprio, G., Jr.; Levine, R. Bank regulation and supervision: What works best? *J. Financ. Intermed.* **2004**, *13*, 205–248. [CrossRef]
- 27. Berle, A.A.; Means, G.C. The Modern Corporation and Private Property; Mac-Millian: New York, NY, USA, 1932.
- 28. Alchian, A.A.; Demsetz, H. Production, information costs, and economic organization. Am. Econ. Rev. 1972, 62, 777–795.
- 29. Jensen, M.C.; Meckling, W.H. Theory of the firm: Managerial behavior, agency costs and ownership structure. *J. Financ. Econ.* **1976**, *3*, 305–360. [CrossRef]
- 30. Fama, E.F.; Jensen, M.C. Agency problems and residual claims. J. Law Econ. 1983, 26, 327–349. [CrossRef]
- 31. Freeman, R.E. Strategic Management: A Stakeholder Approach; Pitman Publishing Inc.: Lanham, MD, USA, 1984.
- 32. Dowling, J.; Pfeffer, J. Organizational legitimacy: Social values and organizational behavior. *Pac. Sociol. Rev.* **1975**, *18*, 122–136. [CrossRef]
- 33. Deegan, C. Environmental disclosures and share prices—A discussion about efforts to study this relationship. In *Accounting Forum;* Taylor Francis: Milton, UK, 2004; Volume 28, pp. 87–97.
- 34. Donaldson, L.; Davis, J.H. Stewardship theory or agency theory: CEO governance and shareholder returns. *Aust. J. Manag.* **1991**, *16*, 49–64. [CrossRef]
- 35. Aldrich, H.E.; Pfeffer, J. Environments of organizations. Annu. Rev. Sociol. 1976, 2, 79–105. [CrossRef]
- Salancik, G.R.; Pfeffer, J. A social information processing approach to job attitudes and task design. *Adm. Sci. Q.* 1978, 23, 224–253. [CrossRef]
- 37. Hillman, A.J.; Cannella, A.A.; Paetzold, R.L. The resource dependence role of corporate directors: Strategic adaptation of board composition in response to environmental change. *J. Manag. Stud.* 2000, *37*, 235–256. [CrossRef]
- 38. Spence, M. Competitive and optimal responses to signals: An analysis of efficiency and distribution. *J. Econ. Theory* **1974**, *7*, 296–332. [CrossRef]
- 39. Ross, S.A. The determination of financial structure: The incentive-signalling approach. Bell J. Econ. 1977, 8, 23–40. [CrossRef]
- 40. Guthrie, J.; Parker, L.D. Corporate social disclosure practice: A comparative international analysis. *Adv. Public Interest Account.* **1990**, *3*, 159–175.

- 41. Cheng, E.C.; Courtenay, S.M. Board composition, regulatory regime and voluntary disclosure. *Int. J. Account.* **2006**, *41*, 262–289. [CrossRef]
- 42. Said, R.; Hj Zainuddin, Y.; Haron, H. The relationship between corporate social responsibility disclosure and corporate governance characteristics in Malaysian public listed companies. *Soc. Responsib. J.* **2009**, *5*, 212–226. [CrossRef]
- 43. Khan, H.U.Z. The effect of corporate governance elements on corporate social responsibility (CSR) reporting: Empirical evidence from private commercial banks of Bangladesh. *Int. J. Law Manag.* **2010**, *52*, 82–109. [CrossRef]
- 44. De Villiers, C.; Naiker, V.; Van Staden, C.J. The effect of board characteristics on firm environmental performance. *J. Manag.* 2011, 37, 1636–1663. [CrossRef]
- 45. Kathy Rao, K.; Tilt, C.A.; Lester, L.H. Corporate governance and environmental reporting: An Australian study. *Corp. Gov. Int. J. Bus. Soc.* **2012**, *12*, 143–163. [CrossRef]
- Lau, C.; Lu, Y.; Liang, Q. Corporate social responsibility in China: A corporate governance approach. J. Bus. Ethics 2014, 119, 1–15. [CrossRef]
- 47. Giannarakis, G. The determinants influencing the extent of CSR disclosure. Int. J. Law Manag. 2014, 56, 393–416. [CrossRef]
- 48. Jizi, M.I.; Salama, A.; Dixon, R.; Stratling, R. Corporate governance and corporate social responsibility disclosure: Evidence from the US banking sector. *J. Bus. Ethics* **2014**, *125*, 601–615. [CrossRef]
- 49. Ibrahim, A.H.; Hanefah, M.M. Board diversity and corporate social responsibility in Jordan. *J. Financ. Report. Account.* **2016**, 14, 279–298. [CrossRef]
- 50. Birindelli, G.; Dell'Atti, S.; Iannuzzi, A.P.; Savioli, M. Composition and activity of the board of directors: Impact on ESG performance in the banking system. *Sustainability* **2018**, *10*, 4699. [CrossRef]
- 51. Rustam, A.; Wang, Y.; Zameer, H. Does foreign ownership affect corporate sustainability disclosure in Pakistan? A sequential mixed methods approach. *Environ. Sci. Pollut. Res.* **2019**, *26*, 31178–31197. [CrossRef]
- 52. Matuszak, Ł.; Różańska, E.; Macuda, M. The impact of corporate governance characteristics on banks' corporate social responsibility disclosure: Evidence from Poland. *J. Account. Emerg. Econ.* **2019**, *9*, 75–102. [CrossRef]
- 53. Aslam, S.; Makki, M.A.M.; Mahmood, S.; Amin, S. Gender diversity and managerial ownership response to corporate social responsibility initiatives: Empirical Evidence from Australia. *J. Manag. Sci.* **2019**, *12*, 131–151.
- 54. Husted, B.W.; de Sousa-Filho, J.M. Board structure and environmental, social, and governance disclosure in Latin America. *J. Bus. Res.* **2019**, *102*, 220–227. [CrossRef]
- 55. Beji, R.; Yousfi, O.; Loukil, N.; Omri, A. Board diversity and corporate social responsibility: Empirical evidence from France. J. Bus. Ethics **2021**, 173, 133–155. [CrossRef]
- 56. Setiawan, D.; Brahmana, R.K.; Asrihapsari, A.; Maisaroh, S. Does a foreign board improve corporate social responsibility? *Sustainability* **2021**, *13*, 11473. [CrossRef]
- 57. Garanina, T.; Aray, Y. Enhancing CSR disclosure through foreign ownership, foreign board members, and cross-listing: Does it work in Russian context? *Emerg. Mark. Rev.* **2021**, *46*, 100754. [CrossRef]
- 58. Ben Fatma, H.; Chouaibi, J. Corporate governance and CSR disclosure: Evidence from European financial institutions. *Int. J. Discl. Gov.* **2021**, *18*, 1–16. [CrossRef]
- 59. Pinheiro, A.B.; dos Santos, J.I.A.S.; Cherobim, A.P.M.S.; Segatto, A.P. What drives environmental, social and governance (ESG) performance? The role of institutional quality. *Manag. Environ. Qual. Int. J.* **2023**, *35*, 427–444. [CrossRef]
- 60. Menicucci, E.; Paolucci, G. The influence of Italian board characteristics on environmental, social and governance dimensions. *Manag. Decis.* **2023**, *61*, 3082–3105. [CrossRef]
- 61. Al Maeeni, F.; Ellili, N.O.D.; Nobanee, H. Impact of corporate governance on corporate social responsibility disclosure of the UAE listed banks. *J. Financ. Report. Account.* 2024, 22, 769–795. [CrossRef]
- 62. Pinheiro, A.B.; Behm, A.J.B.; do Prado, N.B.; Mazzioni, S. The impact of board composition on ESG performance: Comparing results from symmetrical and asymmetrical approaches. *Bus. Strategy Dev.* **2024**, *7*, e424. [CrossRef]
- 63. Abdelazim, A.; Khalaf, B.A. Do Board Characteristics Affect ESG Performance for European Banks? In *The AI Revolution: Driving Business Innovation and Research: Volume 2;* Springer Nature: Cham, Switzerland, 2024; pp. 1035–1052.
- 64. Ntim, C.G.; Soobaroyen, T. Corporate governance and performance in socially responsible corporations: New empirical insights from a Neo-Institutional framework. *Corp. Gov. Int. Rev.* **2013**, *21*, 468–494. [CrossRef]
- 65. Treepongkaruna, S.; Kyaw, K.; Jiraporn, P. ESG controversies and corporate governance: Evidence from board size. *Bus. Strategy Environ.* **2024**, *33*, 4218–4232. [CrossRef]
- 66. Jensen, M.C. The modern industrial revolution, exit, and the failure of internal control systems. *J. Financ.* **1993**, *48*, 831–880. [CrossRef]
- 67. Cheng, S. Board size and the variability of corporate performance. J. Financ. Econ. 2008, 87, 157–176. [CrossRef]
- 68. Tibiletti, V.; Marchini, P.L.; Furlotti, K.; Medioli, A. Does corporate governance matter in corporate social responsibility disclosure? Evidence from Italy in the "era of sustainability". *Corp. Soc. Responsib. Environ. Manag.* **2021**, *28*, 896–907. [CrossRef]
- 69. Miranda, B.; Delgado, C.; Branco, M.C. Board Characteristics, Social Trust and ESG Performance in the European Banking Sector. *J. Risk Financ. Manag.* **2023**, *16*, 244. [CrossRef]
- 70. Zhu, N.; Aryee, E.N.T.; Agyemang, A.O.; Wiredu, I.; Zakari, A.; Agbadzidah, S.Y. Addressing environment, social and governance (ESG) investment in China: Does board composition and financing decision matter? *Heliyon* **2024**, *10*, e30783. [CrossRef] [PubMed]

- 71. Sepulveda-Nuñez, M.D.D.C.; Fong Reynoso, C.; Llamosas-Rosas, I. Board of directors effect on environmental, social and governance performance in publicly traded non-financial firms. *Corp. Gov. Int. J. Bus. Soc.* 2024. [CrossRef]
- 72. Adams, R.B.; de Haan, J.; Terjesen, S.; van Ees, H. Board diversity: Moving the field forward. *Corp. Gov. Int. Rev.* 2015, 23, 77–82. [CrossRef]
- 73. Zhang, J.Q.; Zhu, H.; Ding, H.B. Board composition and corporate social responsibility: An empirical investigation in the post Sarbanes-Oxley era. *J. Bus. Ethics* **2013**, *114*, 381–392. [CrossRef]
- 74. Daniel, N.U.; Urhoghide, R.O. Corporate governance and corporate social responsibility disclosure in Nigerian financial sector. *Int. Account. Tax. Res. Group* **2018**, *2*, 1–13.
- 75. Shakil, M.H.; Tasnia, M.; Mostafiz, M.I. Board gender diversity and environmental, social and governance performance of US banks: Moderating role of environmental, social and corporate governance controversies. *Int. J. Bank Mark.* **2021**, *39*, 661–677. [CrossRef]
- Galletta, S.; Mazzù, S.; Naciti, V.; Vermiglio, C. Gender diversity and sustainability performance in the banking industry. *Corp. Soc. Responsib. Environ. Manag.* 2022, 29, 161–174. [CrossRef]
- Lu, J.; Wang, J. Corporate governance, law, culture, environmental performance and CSR disclosure: A global perspective. J. Int. Financ. Mark. Inst. Money 2021, 70, 101264. [CrossRef]
- 78. Lavin, J.F.; Montecinos-Pearce, A.A. ESG disclosure in an emerging market: An empirical analysis of the influence of board characteristics and ownership structure. *Sustainability* **2021**, *13*, 10498. [CrossRef]
- 79. Nicolò, G.; Zampone, G.; Sannino, G.; De Iorio, S. Sustainable corporate governance and non-financial disclosure in Europe: Does the gender diversity matter? *J. Appl. Account. Res.* **2021**, *23*, 227–249. [CrossRef]
- 80. Yadav, P.; Prashar, A. Board gender diversity: Implications for environment, social, and governance (ESG) performance of Indian firms. *Int. J. Product. Perform. Manag.* 2023, 72, 2654–2673. [CrossRef]
- 81. Wasiuzzaman, S.; Subramaniam, V. Board gender diversity and environmental, social and governance (ESG) disclosure: Is it different for developed and developing nations? *Corp. Soc. Responsib. Environ. Manag.* 2023, 30, 2145–2165. [CrossRef]
- 82. Harjoto, M.; Laksmana, I.; Lee, R. Board diversity and corporate social responsibility. J. Bus. Ethics 2015, 132, 641-660. [CrossRef]
- 83. Muttakin, M.B.; Khan, A.; Subramaniam, N. Firm characteristics, board diversity and corporate social responsibility: Evidence from Bangladesh. *Pac. Account. Rev.* 2015, 27, 353–372. [CrossRef]
- 84. Cucari, N.; Esposito De Falco, S.; Orlando, B. Diversity of board of directors and environmental social governance: Evidence from Italian listed companies. *Corp. Soc. Responsib. Environ. Manag.* **2018**, *25*, 250–266. [CrossRef]
- 85. Issa, A.; Zaid, M.A.A.; Hanaysha, J.R.; Gull, A.A. An examination of board diversity and corporate social responsibility disclosure: Evidence from banking sector in the Arabian Gulf countries. *Int. J. Account. Inf. Manag.* **2022**, *30*, 22–46. [CrossRef]
- Ananzeh, H. Corporate governance and the quality of CSR disclosure: Lessons from an emerging economy. Soc. Bus. Rev. 2022, 17, 280–306. [CrossRef]
- Deschênes, S.; Rojas, M.; Boubacar, H.; Prud'homme, B.; Ouedraogo, A. The impact of board traits on the social performance of Canadian firms. *Corp. Gov.* 2015, 15, 293–305. [CrossRef]
- Oxelheim, L.; Gregorič, A.; Randøy, T.; Thomsen, S. On the internationalization of corporate boards: The case of Nordic firms. J. Int. Bus. Stud. 2013, 44, 173–194. [CrossRef]
- 89. Oxelheim, L.; Randøy, T. The impact of foreign board membership on firm value. J. Bank. Financ. 2003, 27, 2369–2392. [CrossRef]
- 90. Ruigrok, W.; Peck, S.; Tacheva, S. Nationality and gender diversity on Swiss corporate boards. *Corp. Gov. Int. Rev.* 2007, 15, 546–557. [CrossRef]
- 91. Toumi, N.B.F.; Khemiri, R.; Makni, Y.F. Board directors' home regions and CSR disclosure: Evidence from France. J. Appl. Account. Res. 2022, 23, 509–539. [CrossRef]
- Masulis, R.W.; Wang, C.; Xie, F. Globalizing the boardroom—The effects of foreign directors on corporate governance and firm performance. J. Account. Econ. 2012, 53, 527–554. [CrossRef]
- 93. Majeed, S.; Aziz, T.; Saleem, S. The effect of corporate governance elements on corporate social responsibility (CSR) disclosure: An empirical evidence from listed companies at KSE Pakistan. *Int. J. Financ. Stud.* **2015**, *3*, 530–556. [CrossRef]
- 94. Liao, L.; Lin, T.; Zhang, Y. Corporate Board and Corporate Social Responsibility Assurance: Evidence from China. *J. Bus. Ethics* **2018**, *150*, 211–225. [CrossRef]
- 95. Barako, D.G.; Brown, A.M. Corporate social reporting and board representation: Evidence from the Kenyan banking sector. *J. Manag. Gov.* **2008**, *12*, 309–324. [CrossRef]
- 96. Crisóstomo, V.L.; Freire, F.D.S. The influence of ownership concentration on firm resource allocations to employee relations, external social actions, and environmental action. *Rev. Bras. Gestão Negócios* **2015**, *17*, 987–1006. [CrossRef]
- 97. Keim, G.D. Managerial behavior and the social responsibility debate: Goals versus constraints. *Acad. Manag. J.* **1978**, *21*, 57–68. [CrossRef]
- 98. Ullmann, A.A. Data in search of a theory: A critical examination of the relationships among social performance, social disclosure, and economic performance of US firms. *Acad. Manag. Rev.* **1985**, *10*, 540–557. [CrossRef]
- 99. Brammer, S.; Millington, A. Corporate reputation and philanthropy: An empirical analysis. *J. Bus. Ethics* 2005, *61*, 29–44. [CrossRef]
- 100. Li, W.; Zhang, R. Corporate social responsibility, ownership structure, and political interference: Evidence from China. *J. Bus. Ethics* **2010**, *96*, 631–645. [CrossRef]

- López-Iturriaga, F.J.; López-de-Foronda, Ó. Corporate social responsibility and reference shareholders: An analysis of European multinational firms. *Transnatl. Corp. Rev.* 2011, *3*, 17–33. [CrossRef]
- 102. Dam, L.; Scholtens, B. Ownership concentration and CSR policy of European multinational enterprises. *J. Bus. Ethics* **2013**, *118*, 117–126. [CrossRef]
- 103. Rees, W.; Rodionova, T. What type of controlling investors impact on which elements of corporate social responsibility? *J. Sustain. Financ. Investig.* **2013**, *3*, 238–263. [CrossRef]
- 104. Eng, L.L.; Mak, Y.T. Corporate governance and voluntary disclosure. J. Account. Public Policy 2003, 22, 325–345. [CrossRef]
- 105. Harjoto, M.A.; Jo, H. Corporate governance and CSR nexus. J. Bus. Ethics 2011, 100, 45-67. [CrossRef]
- 106. Ho, P.L.; Taylor, G. Corporate governance and different types of voluntary disclosure: Evidence from Malaysian listed firms. *Pac. Account. Rev.* **2013**, *25*, 4–29. [CrossRef]
- Chung, C.Y.; Cho, S.J.; Ryu, D.; Ryu, D. Institutional blockholders and corporate social responsibility. *Asian Bus. Manag.* 2019, 18, 143–186. [CrossRef]
- 108. Kim, I.J.; Eppler-Kim, J.; Kim, W.S.; Byun, S.J. Foreign investors and corporate governance in Korea. *Pac. Basin Financ. J.* 2010, *18*, 390–402. [CrossRef]
- 109. Setiawan, D.; Bandi, B.; Phua, L.K.; Trinugroho, I. Ownership structure and dividend policy in Indonesia. *J. Asia Bus. Stud.* 2016, 10, 230–252. [CrossRef]
- 110. Vo, X.V.; Chu, T.K.H. Do foreign shareholders improve corporate earnings quality in emerging markets? Evidence from Vietnam. *Cogent Econ. Financ.* **2019**, *7*, 1698940. [CrossRef]
- 111. Khan, A.; Muttakin, M.B.; Siddiqui, J. Corporate governance and corporate social responsibility disclosures: Evidence from an emerging economy. *J. Bus. Ethics* **2013**, *114*, 207–223. [CrossRef]
- 112. Oh, W.Y.; Chang, Y.K.; Martynov, A. The effect of ownership structure on corporate social responsibility: Empirical evidence from Korea. J. Bus. Ethics 2011, 104, 283–297. [CrossRef]
- 113. Masud, M.A.K.; Nurunnabi, M.; Bae, S.M. The effects of corporate governance on environmental sustainability reporting: Empirical evidence from South Asian countries. *Asian J. Sustain. Soc. Responsib.* **2018**, *3*, 3. [CrossRef]
- 114. Delgado-Márquez, B.L.; Pedauga, L.E.; Cordón-Pozo, E. Industries regulation and firm environmental disclosure: A stakeholders' perspective on the importance of legitimation and international activities. *Organ. Environ.* **2015**, *30*, 103–121. [CrossRef]
- 115. Martínez-Ferrero, J.; García-Sánchez, I.M. Sustainability assurance and assurance providers: Corporate governance determinants in stakeholder-oriented countries. J. Manag. Organ. 2017, 23, 647–670. [CrossRef]
- 116. Rossi, A.; Tarquinio, L. An analysis of sustainability report assurance statements: Evidence from Italian listed companies. *Manag. Audit. J.* **2017**, *32*, 578–602. [CrossRef]
- 117. Ching, H.Y.; Gerab, F. Sustainability reports in Brazil through the lens of signaling, legitimacy and stakeholder theories. *Soc. Responsib. J.* **2017**, *13*, 95–110. [CrossRef]
- 118. Jain, R.; Winner, L.H. CSR and sustainability reporting practices of top companies in India. *Corp. Commun. Int. J.* **2016**, *21*, 36–55. [CrossRef]
- 119. Bae, S.M.; Masud, M.A.K.; Kim, J.D. A cross-country investigation of corporate governance and corporate sustainability disclosure: A signaling theory perspective. *Sustainability* **2018**, *10*, 2611. [CrossRef]
- 120. Kouser, R.; Bano, T.; Azeem, M.; Ul Hassan, M. Inter-relationship between profitability, growth and size: A case of non-financial companies from Pakistan. *Pak. J. Commer. Soc. Sci.* (*PJCSS*) **2012**, *6*, 405–419.
- 121. Mishra, A.V. Foreign ownership and firm value: Evidence from Australian firms. *Asia Pac. Financ. Mark.* 2014, 21, 67–96. [CrossRef]
- Fuadah, L.L.; Mukhtaruddin, M.; Andriana, I.; Arisman, A. The ownership structure, and the environmental, social, and governance (ESG) disclosure, firm value and firm performance: The audit committee as moderating variable. *Economies* 2022, 10, 314. [CrossRef]
- 123. Ahmed, R.; Abweny, M.; Benjasak, C.; Nguyen, D.T. Financial sanctions and environmental, social, and governance (ESG) performance: A comparative study of ownership responses in the Chinese context. *J. Environ. Manag.* 2024, 351, 119718. [CrossRef] [PubMed]
- 124. Gulzar, M.A.; Cherian, J.; Hwang, J.; Jiang, Y.; Sial, M.S. The impact of board gender diversity and foreign institutional investors on the corporate social responsibility (CSR) engagement of Chinese listed companies. *Sustainability* **2019**, *11*, 307. [CrossRef]
- 125. Al-Gamrh, B.; Al-Dhamari, R.; Jalan, A.; Afshar Jahanshahi, A. The impact of board independence and foreign ownership on financial and social performance of firms: Evidence from the UAE. *J. Appl. Account. Res.* **2020**, *21*, 201–229. [CrossRef]
- 126. Haladu, A.; Beri, M.H. Corporate characteristics and sustainability reporting environmental agencies' moderating effects. *IOSR J. Humanit. Soc. Sci.* **2016**, *21*, 19–30. [CrossRef]
- 127. Ntim, C.G.; Lindop, S.; Thomas, D.A. Corporate governance and risk reporting in South Africa: A study of corporate risk disclosures in the pre-and post-2007/2008 global financial crisis periods. *Int. Rev. Financ. Anal.* 2013, 30, 363–383. [CrossRef]
- Nazari, J.A.; Herremans, I.M.; Warsame, H.A. Sustainability reporting: External motivators and internal facilitators. *Corp. Gov.* 2015, 15, 375–390. [CrossRef]
- 129. Fortune. What Activist Investors Want; 1999; Volume 127, pp. 59-63.
- 130. Holderness, C.G.; Sheehan, D.P. The role of majority shareholders in publicly held corporations: An exploratory analysis. *J. Financ. Econ.* **1988**, *20*, 317–346. [CrossRef]

- 131. Pound, J. Proxy contests and the efficiency of shareholder oversight. J. Financ. Econ. 1988, 20, 237–265. [CrossRef]
- 132. Smith, M.P. Shareholder activism by institutional investors: Evidence from CalPERS. J. Financ. 1996, 51, 227–252. [CrossRef]
- 133. Mahoney, L.; Roberts, R.W. Corporate social performance, financial performance and institutional ownership in Canadian firms. In *Accounting Forum*; Elsevier: Amsterdam, The Netherlands, 2007; Volume 31, pp. 233–253.
- 134. Graves, S.B.; Waddock, S.A. Institutional owners and corporate social performance. *Acad. Manag. J.* **1994**, *37*, 1034–1046. [CrossRef]
- 135. Ahmed, S.U.; Islam, Z.; Mahtab, H.; Hasan, I. Institutional investment and corporate social performance: Linkage towards sustainable development. *Corp. Soc. Responsib. Environ. Manag.* 2014, 21, 1–13. [CrossRef]
- 136. Chen, T.; Dong, H.; Lin, C. Institutional shareholders and corporate social responsibility. *J. Financ. Econ.* **2020**, *135*, 483–504. [CrossRef]
- 137. Kordsachia, O.; Focke, M.; Velte, P. Do sustainable institutional investors contribute to firms' environmental performance? Empirical evidence from Europe. *Rev. Manag. Sci.* 2022, *16*, 1409–1436. [CrossRef]
- 138. Potharla, S.; Mahapatra, S.K.; Turubilli, S.K. Is Institutional Ownership Socially Responsible? Perspectives from Heterogeneity and Stability of Institutional Ownership. *Glob. Bus. Rev.* 2023. [CrossRef]
- 139. Fauzi, H.; Mahoney, L.S.; Abdul Rahman, A. Institutional ownership and corporate social performance: Empirical evidence from Indonesian companies. *Issues Soc. Environ. Account.* **2007**, *1*, 334–347. [CrossRef]
- 140. Coffey, B.S.; Fryxell, G.E. Institutional ownership of stock and dimensions of corporate social performance: An empirical examination. *J. Bus. Ethics* **1991**, *10*, 437–444. [CrossRef]
- 141. Yadav, S. Institutional ownership and corporate social performance in emerging economies multinationals: Evidence from India. *Indian J. Corp. Gov.* **2020**, *13*, 227–252. [CrossRef]
- 142. Acar, E.; Tunca Çalıyurt, K.; Zengin-Karaibrahimoglu, Y. Does ownership type affect environmental disclosure? *Int. J. Clim. Chang. Strateg. Manag.* 2021, *13*, 120–141. [CrossRef]
- 143. Pfeffer, J. Size, Composition and Function on Hospital Board of Directors: A study of Organization-Environmental Linkage. *Adm. Sci. Q.* **1973**, *18*, 349–356. [CrossRef]
- Dalton, R.D.; Daily, M.C.; Johnson, L.J.; Ellstarnd, A.E. Number of Directors and Financial Performance: A Meta-Analysis. Acad. Manag. J. 1999, 42, 674–686. [CrossRef]
- 145. Larmou, S.; Vafeas, N. The Relation Between Board Size and Firm Performance in Firms with a History of Poor Operating Performance. J. Manag. Gov. 2009, 14, 61–85. [CrossRef]
- 146. La Porta, R.; Lopez-De-Silanes, F.; Shleifer, A. Corporate Ownership Around the World. J. Financ. 1999, 54, 471–517. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.